

# G OPEN ACCESS

**Citation:** Akhter Z, Rankin J, Ceulemans D, Ngongalah L, Ackroyd R, Devlieger R, et al. (2019) Pregnancy after bariatric surgery and adverse perinatal outcomes: A systematic review and metaanalysis. PLoS Med 16(8): e1002866. <u>https://doi.</u> org/10.1371/journal.pmed.1002866

Academic Editor: Kathleen Rasmussen, Cornell University, UNITED STATES

Received: March 14, 2019

Accepted: June 28, 2019

Published: August 6, 2019

**Copyright:** © 2019 Akhter et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** All relevant data are within the manuscript and its Supporting Information files.

**Funding:** This study was conducted as part of a Newcastle University Research Excellence Academy PhD Studentship received by ZA (<u>https://</u><u>www.ncl.ac.uk/</u>). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

**RESEARCH ARTICLE** 

# Pregnancy after bariatric surgery and adverse perinatal outcomes: A systematic review and meta-analysis

Zainab Akhter<sup>1\*</sup>, Judith Rankin<sup>1</sup>, Dries Ceulemans<sup>2</sup>, Lem Ngongalah<sup>1</sup>, Roger Ackroyd<sup>3</sup>, Roland Devlieger<sup>2</sup>, Rute Vieira<sup>4</sup>, Nicola Heslehurst<sup>1</sup>

1 Institute of Health & Society, Newcastle University, Newcastle upon Tyne, United Kingdom, 2 Department of Obstetrics and Gynaecology, University Hospitals Leuven, Leuven, Belgium, 3 Department of Surgery, Sheffield Teaching Hospitals, Sheffield, United Kingdom, 4 Institute of Health Sciences Research, University of Aberdeen, Aberdeen, United Kingdom

\* z.akhter@newcastle.ac.uk

# Abstract

# Background

Women who undergo bariatric surgery prior to pregnancy are less likely to experience comorbidities associated with obesity such as gestational diabetes and hypertension. However, bariatric surgery, particularly malabsorptive procedures, can make patients susceptible to deficiencies in nutrients that are essential for healthy fetal development. The objective of this systematic review and meta-analysis is to investigate the association between pregnancy after bariatric surgery and adverse perinatal outcomes.

# Methods and findings

Searches were conducted in Medline, Embase, PsycINFO, CINAHL, Scopus, and Google Scholar from inception to June 2019, supplemented by hand-searching reference lists, citations, and journals. Observational studies comparing perinatal outcomes post-bariatric surgery to pregnancies without prior bariatric surgery were included. Outcomes of interest were perinatal mortality, congenital anomalies, preterm birth, postterm birth, small and large for gestational age (SGA/LGA), and neonatal intensive care unit (NICU) admission. Pooled effect sizes were calculated using random-effects meta-analysis. Where data were available, results were subgrouped by type of bariatric surgery. We included 33 studies with 14,880 pregnancies post-bariatric surgery and 3,979,978 controls. Odds ratios (ORs) were increased after bariatric surgery (all types combined) for perinatal mortality (1.38, 95% confidence interval [CI] 1.03–1.85, p = 0.031), congenital anomalies (1.29, 95% CI 1.04–1.59, p = 0.019), preterm birth (1.57, 95% CI 1.38–1.79, p < 0.001), and NICU admission (1.41, 95% CI 1.25–1.59, p < 0.001). Postterm birth decreased after bariatric surgery (OR 0.46, 95% CI 0.35–0.60, p < 0.001). ORs for SGA increased (2.72, 95% CI 2.32–3.20, p < 0.001) and LGA decreased (0.24, 95% CI 0.14–0.41, p < 0.001) after gastric bypass but not after gastric banding. Babies born after bariatric surgery (all types combined) weighed over 200 g less than those born to mothers without prior bariatric surgery (weighted mean difference



Abbreviations: BMI, body mass index; BPD, biliopancreatic diversion; CI, confidence interval; GWG, gestational weight gain; IPD, individual patient data; LAGB, laparoscopic adjustable gastric banding; LGA, large for gestational age; NICU, neonatal intensive care unit; NR, not reported; OR, odds ratio; RYGB, Roux-en-Y gastric bypass; SG, sleeve gastrectomy; SGA, small for gestational age; WMD, weighted mean difference. -242.42 g, 95% CI -307.43 to -177.40 g, p < 0.001). There was low heterogeneity for all outcomes ( $l^2 < 40\%$ ) except LGA. Limitations of our study are that as a meta-analysis of existing studies, the results are limited by the quality of the included studies and available data, unmeasured confounders, and the small number of studies for some outcomes.

### Conclusions

In our systematic review of observational studies, we found that bariatric surgery, especially gastric bypass, prior to pregnancy was associated with increased risk of some adverse perinatal outcomes. This suggests that women who have undergone bariatric surgery may benefit from specific preconception and pregnancy nutritional support and increased monitoring of fetal growth and development. Future studies should explore whether restrictive surgery results in better perinatal outcomes, compared to malabsorptive surgery, without compromising maternal outcomes. If so, these may be the preferred surgery for women of reproductive age.

# **Trial registration**

PROSPERO CRD42017051537.

# Author summary

### Why was this study done?

- Obesity during pregnancy increases the risk of health complications for both mother and baby.
- Bariatric surgery before pregnancy improves obesity-related problems for the mother but reduces the absorption of micronutrients that are needed for healthy fetal development.
- This research aimed to investigate whether bariatric surgery is associated with adverse outcomes for the baby.

# What did the researchers do and find?

- This systematic review included 33 studies that investigated perinatal outcomes among women with previous bariatric surgery compared to women without previous bariatric surgery.
- Meta-analysis identified a significant increase in odds of perinatal mortality, congenital anomalies, preterm birth, and neonatal intensive care unit admission but a decrease in odds of postterm birth after bariatric surgery.
- The odds of small babies were increased and the odds of large babies were decreased after malabsorptive bariatric surgery types, but there was no change for restrictive bariatric surgery types.

#### What do these findings mean?

- Bariatric surgery, in particular malabsorptive types of surgery, seems associated with an increased risk of some adverse perinatal outcomes, which suggests a link with nutrition.
- Women of reproductive age undergoing bariatric surgery are a high-risk group and require specialised preconception and antenatal nutritional support to achieve the best outcomes for both mothers and babies.

# Introduction

Obesity is a global public health challenge with over 650 million adults affected worldwide, and prevalence continues to rise, making obesity the most common medical condition in women of reproductive age [1,2]. Maternal obesity, defined as prepregnancy body mass index (BMI)  $\geq$  30 kg/m<sup>2</sup>, has severe implications for both mother and baby. Maternal risks include higher likelihood of gestational diabetes, preeclampsia, and cesarean section [3]. For the neonate, there is increased risk of pre- and postterm birth, small and large for gestational (SGA/ LGA), congenital anomalies, and perinatal mortality [3,4]. Interventions to reduce maternal obesity are important not only to improve pregnancy outcomes but also to reduce the longterm health burden on the mother and offspring, including cardiovascular disease and insulin resistance [5].

Bariatric surgery is the most effective treatment for long-term weight loss, and over half of surgeries are performed on women of reproductive age [6,7]. Women who undergo bariatric surgery prior to pregnancy are less likely to experience comorbidities associated with obesity, such as gestational diabetes and hypertension [8]. However, micronutrient deficiencies are increased after bariatric surgery and may therefore have implications for fetal environment [9]. Maternal deficiencies in folate, iron, and vitamin D, for example, are all linked with adverse perinatal outcomes including neural tube defects, preterm birth, and low birth weight [10]. Malabsorptive procedures such as Roux-en-Y gastric bypass (RYGB) and bilio-pancreatic diversion (BPD) reduce the absorption of micronutrients because part of the small intestine is bypassed, whereas restrictive procedures such as laparoscopic adjustable gastric banding (LAGB) and sleeve gastrectomy (SG) reduce stomach capacity [11]. There have been multiple case reports of congenital anomalies occurring after malabsorptive procedures because of maternal malnutrition; however, the evidence from observational studies is conflicting [12].

Previous meta-analyses on pregnancy after bariatric surgery have focused on maternal outcomes, and there is limited evidence on perinatal outcomes other than size for gestational age and preterm birth [8,13,14]. The aim of this systematic review and meta-analysis was to compare adverse perinatal outcomes among women who underwent bariatric surgery prior to pregnancy with those who had not. When possible, the difference in effect size between malabsorptive and restrictive procedures was explored.

### Methods

#### Search strategy and selection criteria

Searches were conducted in Medline, Embase, PsycINFO, CINAHL, Scopus, Google Scholar, and relevant e-journals from inception to June 3, 2019. We included observational studies published in the English language, involving women who had undergone bariatric surgery prior to pregnancy, and compared them to women without a history of bariatric surgery. We included studies that combined all types of bariatric surgery or provided data for RYGB, LAGB, SG, or BPD separately. The following perinatal outcomes were included: perinatal mortality (including stillbirth), congenital anomalies, preterm birth, postterm birth, SGA, LGA, neonatal intensive care unit (NICU) admission, birth weight, and gestational age.

The search strategy (<u>S1 Table</u>) included a mixture of keywords and MeSH headings: (pregnan\* or mother\* or matern\*) and (bariatric surgery or weight loss and surgery or gastric bypass or gastric band\* or sleeve or biliopancreatic diversion or LAGB or RYGB) and (death or mortality or newborn\* or fetal or congenital or stillbirth or defect\* or perinatal or obstetric or neonat\* or outcome\* or birth). Reference lists and citations were searched for all included primary studies and for relevant reviews identified by the database searches. Authors were contacted if additional data were required for inclusion in meta-analysis. Screening, data extraction, and quality assessment were carried out in duplicate.

This review was conducted in line with the PRISMA and MOOSE guidelines (<u>S1 PRISMA</u> <u>Checklist</u>) [<u>15,16</u>]. The protocol is published on PROSPERO (CRD42017051537).

#### Data analysis

The Cochrane Cohort Study data extraction tool was adapted to meet the requirements of this review. Study characteristics extracted included study design, study location, type of bariatric surgery, and control group. Frequencies, effect sizes, and confidence intervals (CIs) of adverse perinatal outcomes were also extracted. For continuous outcomes, means and standard deviations were extracted. When multiple studies reported data from the same cohort with the same participant inclusion criteria, the decision was made to include the study with the larger sample size for the exposed group. Studies with duplicate data were only included if they reported different perinatal outcomes and were therefore included in separate meta-analysis. The Newcastle-Ottawa quality assessment scale was used to appraise the quality of the included studies out of a maximum of eight points (S1 Fig). The studies were assessed for representativeness of the exposed cohort, selection of the nonexposed cohort, ascertainment of exposure and outcome, study design and analysis, and adequacy of follow-up.

A meta-analysis was used to calculate a pooled odds ratio (OR) and 95% CI when there were at least three studies reporting the same outcome. For continuous perinatal outcomes, a weighted mean difference (WMD) and 95% CI were calculated. DerSimonian and Laird random-effects model was used to take clinical heterogeneity into account such as unreported differences between surgical procedures (e.g., technique and limb length) and different levels of patient postsurgery and preconception care. When a study reported data on multiple control groups, a hierarchy was developed to firstly include the most comparable BMI group to the postbariatric patient, which was prepregnancy BMI matched, then obesity. When there was evidence of moderate heterogeneity ( $I^2 > 40\%$ ), subgroup analysis by type of surgery or comparison group, as defined a priori, was carried out if three or more studies existed for each group. Any remaining heterogeneity was explored through meta-regression for factors including location, sample size, publication date, and quality. Publication bias was investigated using Egger's test and funnel plots. For studies reporting adjusted results, their crude and adjusted

ORs were compared to determine whether adjustments affected the effect size. Sensitivity analysis was performed for each meta-analysis by excluding one study at a time to identify the effect of any individual study on the pooled effect size and between-study heterogeneity. All analyses were conducted in Stata/SE 15.0.

### Results

#### Study characteristics

Database searches identified 3,470 results for title and abstract screening, of which 141 studies underwent full-text assessment (Fig 1). The kappa statistic for inter-rater agreement of study inclusion between authors was 0.84 (scores > 0.81 are considered excellent) [17]. Thirty-seven studies met the inclusion criteria, but four were excluded because they reported the same cohort, participant inclusion criteria, and outcomes as another study [18-21]. This resulted in 33 studies that reported original data on perinatal outcomes (14,880 pregnancies after bariatric surgery and 3,979,978 pregnancies without bariatric surgery, Table 1). Fifteen of the included studies were conducted in Europe, 10 were conducted in the United States, three in Israel, two in each Australia and Brazil, and one in Canada. Studies were published between 1998 and 2018. All studies scored over five out of eight for quality, with 20 studies scoring at least seven (S2 Table). Many studies conducted more than one analysis with multiple surgery types or control groups. Sixteen analyses combined all bariatric surgery patients, whereas 14 studies were restricted to RYGB, six analyses included only LAGB, one included only SG, and one included BPD. Nine analyses compared women's postsurgical pregnancies to pre/early-pregnancy BMI-matched controls, and 14 used obesity controls (which were > 30 kg/m<sup>2</sup>, 35 kg/  $m^2$ , or 40 kg/m<sup>2</sup>) in line with their relevant bariatric surgery guidelines, or matched for presurgical BMI. Eleven analyses compared pregnancies before and after bariatric surgery, nine compared outcomes to the general population, and five used healthy BMI as the control group.

#### Perinatal mortality and congenital anomalies

Perinatal mortality or stillbirth was reported in 10 studies. The pooled odds were significantly increased post-bariatric surgery compared to women without prior bariatric surgery (OR 1.38, 95% CI 1.03–1.85, p = 0.031) (Fig 2A) [22,25,26,34,39,41–43,45,52]. Ten studies reported on congenital anomalies, which were also found to have significantly increased odds post-bariatric surgery (OR 1.29, 95% CI 1.04–1.59, p = 0.019) (Fig 2B) [22,25,34,35,41,42,44,46,51,52]. There was no significant heterogeneity for either outcome ( $I^2 = 12.1\%$ , 95% CI 0.0–53.1, p = 0.331 and  $I^2 = 28\%$ , 95% CI 0.0–65.5, p = 0.186, respectively).

#### Gestational age

Preterm birth was reported in 20 studies, with 19 eligible for meta-analysis [22,27– 30,32,34,36–39,42,44–47,49,51,53]. The overall odds of preterm birth were significantly increased post-bariatric surgery compared to women without prior bariatric surgery (OR 1.35, 95% CI 1.14–1.60, p = 0.001) (S2 Fig). There was significant heterogeneity ( $I^2 = 50.1\%$ , 95% CI 15.3–70.6, p = 0.007), which remained significant after subgroup analyses by control group but was reduced after subgrouping by type of surgery (Fig 3A). There were significantly increased odds of preterm birth after bariatric surgery in the 'all bariatric surgery' group (OR 1.57, 95% CI 1.38–1.79, p < 0.001). The association was not significant for subgroups 'RYGB' (OR 1.14, 95% CI 0.89–1.46, p = 0.289) or 'LAGB or SG' (OR 0.88, 95% CI 0.58–1.34, p = 0.565). The study excluded from the meta-analysis because of lack of crude data reported an adjusted OR for preterm birth of 1.43 (95% CI 1.01–2.03) post-bariatric surgery (n = 293) compared to



Fig 1. PRISMA flowchart of included studies. BMI, body mass index.

https://doi.org/10.1371/journal.pmed.1002866.g001

general population controls (n = 656,353) [23]. Postterm birth was reported in five studies, and the odds more than halved after bariatric surgery (OR 0.46, 95% CI 0.35–0.60, p < 0.001) (Fig 3B) [22,37,44,50,51]. There was no significant heterogeneity ( $I^2 = 7.2\%$ , 95% CI 0.0–80.7, p = 0.366).

Despite the results of increased preterm birth and decreased postterm birth, the WMD of 13 studies reporting continuous gestational age did not reach statistical significance (WMD -0.16 weeks, 95% CI -0.38 to 0.06, p = 0.156) (S3 Fig) [22-24,30,33,36,38,40,41,44,49,51,52].

Author, publication year, country	Study period	Exposed groups*	Comparison groups*	Perinatal outcomes reported
Adams et al. 2015 [ <u>22],</u> USA	Bariatric surgery between 1979 and 2011	(1) 764 pregnancies after RYGB (2) 2,666 pregnancies after RYGB	<ul><li>(1) 764 pregnancies matched for ppBMI</li><li>(2) 10,447 pregnancies before RYGB</li></ul>	Birth weight Congenital anomalies Gestational age LGA Postterm birth Preterm birth Stillbirth SGA
Belogolovkin et al. 2012 [23], USA	Delivery between 2004 and 2007	293 pregnancies after bariatric surgery	656,353 general population pregnancies	Birth weight Gestational age Macrosomia Preterm birth SGA
Berglind et al. 2014 [ <u>24],</u> Sweden	Bariatric surgery between 1980 and 2006	124 pregnancies after bariatric surgery	124 pregnancies before bariatric surgery	Birth weight Gestational age
Berlac et al. 2014 [ <u>25</u> ], Denmark	Bariatric surgery between January 1996 and June 2011	415 pregnancies after RYGB	827 pregnancies matched for ppBMI 829 healthy BMI 20–24 kg/m <sup>2</sup> pregnancies	Congenital anomalies NICU admission Stillbirth
Burke et al. 2010 [ <u>26],</u> USA	Bariatric surgery between 2002 and 2006	354 pregnancies after bariatric surgery	346 pregnancies matched for presurgery BMI	LGA Stillbirth
Chevrot et al. 2016 [ <u>27</u> ], France	Delivery between January 1, 2004, and December 31, 2013	<ol> <li>(1) 139 pregnancies after bariatric surgery</li> <li>(2) 58 pregnancies after RYGB</li> <li>(2) 81 pregnancies after LAGB or SG</li> </ol>	<ol> <li>(1) 139 pregnancies matched for presurgery BMI</li> <li>(2) 139 pregnancies matched for ppBMI</li> </ol>	Birth weight LGA NICU admission Preterm birth SGA
Dell'Agnolo et al. 2011 [ <u>28],</u> Brazil	Pregnancy between 1999 and 2008	41 pregnancies after bariatric surgery	14 pregnancies before bariatric surgery	Low birth weight Preterm birth
Dixon et al. 2005 [ <u>29</u> ], Australia	Bariatric surgery between January 1, 1995, and August 31, 2003	79 pregnancies after LAGB	79 pregnancies with obesity $> 35 \text{ kg/m}^2$ 40 pregnancies before LAGB 61,000 general population pregnancies	Birth weight Low birth weight Macrosomia Preterm birth
Ducarme et al. 2007 [ <u>30</u> ], France	Delivery between January 2004 and October 2006	13 pregnancies after LAGB	414 pregnancies with obesity > 30kg/m <sup>2</sup>	Gestational age Low birth weight Macrosomia Preterm birth
Feichtinger et al. 2016 [ <u>31</u> ], Austria	Pregnancy between January 2007 and January 2016	76 pregnancies after RYGB	76 pregnancies with obesity > 30 kg/m <sup>2</sup> 76 pregnancies matched for ppBMI 76 healthy BMI 18–25 kg/m <sup>2</sup> pregnancies	LGA NICU admission SGA
Gascoin et al. 2017 [ <u>9],</u> France	Delivery between March 1, 2008, and October 31, 2012	56 pregnancies after RYGB	56 nonobesity pregnancies	Birth weight
Goldman et al. 2016 [ <u>32</u> ], USA	Bariatric surgery between 2002 and 2012	<ol> <li>(1) 12 pregnancies after RYGB</li> <li>(2) 14 pregnancies after LAGB</li> </ol>	<ul> <li>(1)(2) 14 pregnancies with obesity</li> <li>(eligible for bariatric surgery)</li> <li>(1) 36 pregnancies before RYGB</li> <li>(2) 28 pregnancies before LAGB</li> </ul>	Birth weight Preterm birth
Hammeken et al. 2017 [33], Denmark	Delivery between January 1, 2010, and December 31, 2013	151 pregnancies after RYGB	151 pregnancies matched for ppBMI	Birth weight Gestational age LGA NICU admission SGA
Johansson et al. 2015 [ <u>34</u> ], Sweden	Bariatric surgery between 2006 and 2011	596 pregnancies after bariatric surgery	2,356 pregnancies matched for presurgery BMI	Congenital anomalies LGA Preterm birth SGA Stillbirth

#### Table 1. Table of studies included in the systematic review and meta-analysis.

(Continued)

# 

#### Table 1. (Continued)

Author, publication year, country	Study period	Exposed groups*	Comparison groups*	Perinatal outcomes reported
Josefsson et al. 2013 [ <u>35</u> ], Sweden	Mothers born between 1973 and 1983	318 pregnancies after bariatric surgery	244,294 general population pregnancies	Congenital anomalies
Josefsson et al. 2011 [ <u>36</u> ], Sweden	Mothers born between 1973 and 1983	126 pregnancies after bariatric surgery	188,500 general population pregnancies	Birth weight Gestational age LGA Preterm birth SGA
Kjaer et al. 2013 [ <u>37</u> ], Denmark	Delivery between January 2004 and December 2010	<ul><li>(1) 339 pregnancies after bariatric surgery</li><li>(2) 286 pregnancies after RYGB</li></ul>	(1)(2) 1,277 pregnancies matched for ppBMI	LGA Postterm birth Preterm birth SGA
Lapolla et al. 2010 [ <u>38</u> ], Italy	Bariatric surgery between September 1993 and December 2005	<ul><li>(1) 83 pregnancies after LAGB</li><li>(2) 27 pregnancies after LAGB</li></ul>	<ul> <li>(1) 120 pregnancies with obesity &gt; 40 kg/m<sup>2</sup></li> <li>(1) 858 healthy BMI (criteria NR) pregnancies</li> <li>(2) 27 pregnancies before LAGB</li> </ul>	Birth weight Gestational age LGA NICU admission Preterm birth SGA
Lesko and Peaceman 2012 [ <u>39</u> ], USA	Delivery between December 1, 2005, and December 1, 2009	70 pregnancies after bariatric surgery	140 pregnancies matched for presurgery BMI 140 pregnancies matched for ppBMI	Macrosomia NICU admission Preterm birth Stillbirth SGA
Machado et al. 2017 [ <u>40</u> ], Brazil	Pregnancy between March 2008 and March 2012	30 pregnancies after RYGB	60 pregnancies matched for ppBMI	Birth weight Gestational age SGA
Marceau et al. 2004 [ <u>41],</u> Canada	Bariatric surgery before 2000	251 pregnancies after BPD	1,577 pregnancies before BPD	Birth weight Congenital anomalies Gestational age LGA SGA Stillbirth
Parent et al. 2017 [ <u>42</u> ], USA	Delivery between January 1, 1980, and May 30, 2013	1,859 pregnancies after bariatric surgery	8,437 general population pregnancies	Congenital anomalies LGA NICU admission Preterm birth SGA Stillbirth
Parker et al. 2016 [ <u>43</u> ], USA	Delivery in 2012	1,585 pregnancies after bariatric surgery	185,120 pregnancies with obesity $>$ 30 kg/m <sup>2</sup>	LGA SGA Stillbirth
Patel et al. 2008 [44], USA	Delivery between 2003 and 2006	26 pregnancies after RYGB	66 pregnancies with obesity > 30 kg/m <sup>2</sup> 188 nonobesity BMI < 30 kg/ m <sup>2</sup> pregnancies	Birth weight Congenital anomalies Gestational age Macrosomia Postterm birth Preterm birth SGA
Roos et al. 2013 [ <u>45</u> ], Sweden	Delivery between 1992 and 2009	2,534 pregnancies after bariatric surgery	12,468 pregnancies matched for ppBMI 1,740,140 general population pregnancies	LGA Preterm birth SGA Stillbirth
Rottenstreich et al. 2018 [46], Israel	Delivery between 2006 and 2016	119 pregnancies after SG	119 pregnancies matched for presurgery BMI	Congenital anomalies LGA NICU admission Preterm birth SGA

(Continued)

Author, publication year, country	Study period	Exposed groups*	Comparison groups*	Perinatal outcomes reported
Shai et al. 2014 [ <u>47</u> ], Israel	Delivery between 1988 and 2010	326 pregnancies after bariatric surgery	1,612 pregnancies with obesity $>$ 30 kg/ $m^2$	Preterm birth
Skull et al. 2004 [ <u>48],</u> Australia	Bariatric surgery between 1996 and 2003	49 pregnancies after LAGB	31 pregnancies before LAGB	Birth weight
Stentebjerg et al. 2017 [ <u>49</u> ], Denmark	Delivery between November 2007 and October 2013	71 pregnancies after RYGB	57,970 general population pregnancies	Birth weight Gestational age Preterm birth
Stephansson et al. 2018 [50], Sweden	Delivery between 1 January 2006 and 31 December 2013	1,431 pregnancies after bariatric surgery	4,476 pregnancies matched for presurgery BMI 798,338 general population pregnancies	Postterm birth
Wax et al. 2008 [ <u>51</u> ], USA	NR	38 pregnancies after RYGB	76 general population pregnancies	Birth weight Congenital anomalies Gestational age Macrosomia NICU admission Postterm birth Preterm birth SGA
Weintraub et al. 2008 [52], Israel	Delivery between 1988 and 2006	507 pregnancies after bariatric surgery	301 pregnancies before bariatric surgery	Birth weight Congenital anomalies Gestational age IUGR Macrosomia Stillbirth
Wittgrove et al. 1998 [53], USA	NR	36 pregnancies after RYGB	23 pregnancies before RYGB	Macrosomia Preterm birth

#### Table 1. (Continued)

The term 'bariatric surgery' is used when a study combined all types of surgery or did not specify a surgery type.

\*Some studies reported multiple exposed groups and multiple comparison groups. In the case of multiple exposed groups, numbers indicate which comparison group was used. There are no numbers when a single exposed group was compared to all listed comparison groups.

Abbreviations: BMI, body mass index; BPD, biliopancreatic diversion; IUGR, intrauterine growth restriction; LAGB, laparoscopic adjustable gastric banding; LGA, large for gestational age; NICU, neonatal intensive care unit; NR, not reported; ppBMI, prepregnancy BMI; RYGB, Roux-en-Y gastric bypass; SG, sleeve gastrectomy; SGA, small for gestational age.

https://doi.org/10.1371/journal.pmed.1002866.t001

Heterogeneity between studies was substantial and did not reduce with subgroup analyses for type of bariatric surgery. Meta-regression revealed that the following factors did not contribute to heterogeneity: type of surgery, control group, publication year, continent, sample size, or quality score (<u>S3A Table</u>).

#### Size for gestational age and birth weight

SGA, intrauterine growth restriction, and low birth weight were investigated in 22 studies, and 21 of these were eligible for meta-analysis [22,27–31,33,34,36–46,51,52]. The odds of an SGA baby post-bariatric surgery were more than doubled (OR 2.13, 95% CI 1.80–2.52, p < 0.001) (S4 Fig). There was significant evidence of heterogeneity ( $I^2 = 47.0\%$ , 95% CI 11.8–68.2, p = 0.009), which was reduced by subgroup analyses by surgery type (Fig 4A). Odds of SGA were significantly increased for the 'all bariatric surgery' group (OR 1.87, 95% CI 1.61–2.17, p < 0.001) and were further increased for 'RYGB or BPD' (OR 2.72, 95% CI 2.32–3.20, p < 0.001). There was no association between SGA and 'LAGB or SG' (OR 1.25, 95% CI 0.62–2.51, p = 0.533). The study excluded from the meta-analysis reported an adjusted OR of 2.69





**Fig 2. Perinatal mortality and congenital anomalies after bariatric surgery meta-analysis.** Association between maternal bariatric surgery and (A) perinatal mortality (includes stillbirth) and (B) congenital anomalies. Studies are presented as Author, year. The forest plots are stratified by type of surgery. *n* = cases of perinatal mortality or congenital anomalies. *N* = total group size. \*BPD only. BPD, biliopancreatic diversion; CI, confidence interval; OR, odds ratio; ppBMI, prepregnancy body mass index matched; RYGB, Roux-en-Y gastric bypass; SG, sleeve gastrectomy.

https://doi.org/10.1371/journal.pmed.1002866.g002

A – Pre-term birth	Control group	Surgery (n/N)	Control (n/N)	OR (95% CI)
All bariatric surgery				
Dell'Agnolo et al. 2011	Before Surgery	2/41	1/14	0.67 (0.06, 7.97
Josefsson et al. 2011	General Population	11/126	11667/188500	1.45 (0.78, 2.69
Lesko and Peaceman, 2012	ppBMI Matched	14/70	9/140	
Kjaer et al. 2013	ppBMI Matched	29/339	85/1277	1.31 (0.85, 2.04
Roos et al. 2013	ppBMI Matched	243/2511	750/12379	➡ 1.66 (1.43, 1.93)
Shai et al. 2014	Obesity	24/326	110/1612	1.09 (0.69, 1.72
Johansson et al. 2015	Obesity	59/590	176/2344	1.37 (1.00, 1.87
Chevrot et al. 2016	Obesity	12/139	10/139	1.22 (0.51, 2.92
Parent et al. 2017	General Population	261/1859	723/8437	➡ 1.74 (1.50, 2.03
Subtotal (I-squared = 21.0%	, p = 0.256)			1.57 (1.38, 1.79
RYGB				
Wittorove et al. 1998	Before Surgery	4/36	3/23	0.83 (0.17, 4.12
Patel et al. 2008	Obesity	7/26	14/66	1.37 (0.48, 3.90
Wax et al. 2008	General Population	10/38	17/76	1.24 (0.50, 3.05
Kiaer et al. 2013	ppBMI Matched	25/286	74/1070	1.29 (0.80, 2.07
Adams et al. 2015	ppBMI Matched	96/764	108/764	0.87 (0.65, 1.17
Chevrot et al. 2016	ppBMI Matched	6/58	13/139	1.12 (0.40, 3.10
Goldman et al. 2016	Obesity	2/12	2/14	→ 1.20 (0.14, 10.1
Stentebierg et al. 2017	General Population	10/71	3884/57970 -	2.28 (1.17, 4.46
Subtotal (I-squared = 9.8%,	p = 0.354)		►	> 1.14 (0.89, 1.46
LAGB and*/or <sup>†</sup> SG				
Dixon et al. 2005	Obesity	5/79	10/79	- 0.47 (0.15, 1.43
Ducarme et al. 2007	Obesity	1/13	29/414	1.11 (0.14, 8.81
Lapolia et al. 2010	Obesity	15/83	16/120	• 1.43 (0.67, 3.09
Kiaer et al. 2013	ppBMI Matched	4/53	11/207	• 1.45 (0.44, 4.76
Chevrot et al. 2016*	ppBMI Matched	6/81	13/139	0.78 (0.28, 2.13
Goldman et al. 2016	Obesity	1/14	2/14	0.46 (0.04, 5.77
Rottenstreich et al. 2018 <sup>†</sup>	Obesity	8/119	13/119	0.59 (0.23, 1.47
Subtotal (I-squared = 0.0%,	p = 0.599)		$\triangleleft$	0.88 (0.58, 1.34
NOTE: Weights are from ran	dom effects analysis			



**Fig 3. Preterm and postterm birth after bariatric surgery meta-analysis.** Association between maternal bariatric surgery and (A) preterm birth (<37 weeks) and (B) postterm birth (>41 or >42 weeks). Studies are presented as Author, year. The forest plots are stratified by type of surgery, with separate pooled OR (95% CI) when subgroup analysis was possible. *n* = cases of preterm or postterm birth. *N* = total group size. \*LAGB and SG. <sup>†</sup>SG only. CI, confidence interval; LAGB, laparoscopic adjustable gastric banding; OR, odds ratio; ppBMI, prepregnancy body mass index matched; RYGB, Roux-en-Y gastric bypass; SG, sleeve gastrectomy.

https://doi.org/10.1371/journal.pmed.1002866.g003

A - Small for gestational age	Control group	Surgery (n/N)	Control (n/N)	OR (95% CI)
All bariatric surgery		The first Ph		
Weintraub et al. 2008	Before Surgery	20/507	7/301	1.72 (0.72, 4.13)
Dell'Agnolo et al. 2011	Before Surgery	8/41	1/14	◆ 3.15 (0.36.27.76
Josefsson et al. 2011	General Population	7/126	3982/188500	2.73 (1.27, 5.85)
Lesko and Peaceman, 2012	ppBMI Matched	12/70	12/140	2.21 (0.94, 5.21)
Kiaer et al. 2013	ppBMI Matched	24/339	37/1277 -	2.55 (1.51, 4.33)
Roos et al. 2013	ppBMI Matched	131/2507	369/12338	1.79 (1.46, 2.19)
Johansson et al. 2015	Obesity	92/590	178/2336	2 24 (1.71, 2.93)
Chevrot et al. 2016	Obesity	24/139	12/139	2.21 (1.06, 4.62)
Parent et al. 2017	General Population	242/1859	754/8437	1.52 (1.31, 1.78)
Subtotal (I-squared = 23.6%	i, p = 0.234)			> 1.87 (1.61, 2.17)
RYGB or BPD*				
Marceau et al. 2004*	Before Surgery	15/156	20/638 -	3.29 (1.64, 6.58)
Patel et al. 2008	Obesity	3/26	2/66	4.17 (0.66, 26.59
Wax et al. 2008	General Population	3/38	7/76 •	0.84 (0.21, 3.47)
Kjaer et al. 2013	ppBMI Matched	22/286	30/1070 -	2.89 (1.64, 5.09)
Adams et al. 2015	ppBMI Matched	92/764	43/764 -	2.30 (1.57, 3.35)
Chevrot et al. 2016	ppBMI Matched	17/58	8/139	6.79 (2.73, 16.88
Feichtinger et al. 2016	ppBMI Matched	18/64	7/63	3.13 (1.20, 8.14)
Parker et al. 2016	Obesity	90/1585	4073/185120	2.68 (2.16, 3.32)
Hammeken et al. 2017	ppBMI Matched	16/151	6/151	2.86 (1.09, 7.53)
Machado et al. 2017	ppBMI Matched	7/30	5/60	<ul> <li>3.35 (0.96, 11.65</li> </ul>
Subtotal (I-squared = 0.0%)	p = 0.529)		1223 2017	2.72 (2.32, 3.20)
LAGB and <sup>1</sup> /or <sup>2</sup> SG				
Dixon et al. 2005	Obesity	5/79	7/79	0.69 (0.21, 2.29)
Ducarme et al. 2007	Obesity	1/13	44/414 🗲 🔳	0.70 (0.09, 5.52)
Lapolla et al. 2010	Obesity	1/83	5/120 < 🔳	0.28 (0.03, 2.45)
Kjaer et al. 2013	ppBMI Matched	2/53	7/207	1.12 (0.23, 5.56)
Chevrot et al. 2016 <sup>†</sup>	ppBMI Matched	7/81	8/139	1.55 (0.54, 4.44)
Rottenstreich et al. 2018 <sup>1</sup>	Obesity	17/119	5/119	• 3.80 (1.35, 10.67
Subtotal (I-squared = 34.05	a, p = 0.181)			1.25 (0.62, 2.51)
NOTE: Weights are from rar	dom effects analysis			
				<del>1 1 1</del>
			2 5 1	2 5 10



**Fig 4. Size for gestational age after bariatric surgery meta-analysis.** Association between maternal bariatric surgery and (A) small for gestational age (includes low birth weight < 2,500 g for three studies) and (B) large for gestational age (includes macrosomia > 4,000 g for seven studies). Studies are presented as Author, year. Results are subgrouped by type of surgery. *n* = cases of small or large for gestational age. *N* = total group size. \*BPD only. <sup>†</sup>LAGB and SG. <sup>‡</sup>SG only. BPD, biliopancreatic diversion; CI, confidence interval; LAGB, laparoscopic adjustable gastric banding; OR, odds ratio; ppBMI, prepregnancy body mass index matched; RYGB, Roux-en-Y gastric bypass; SG, sleeve gastrectomy.

https://doi.org/10.1371/journal.pmed.1002866.g004

(95% CI 1.96–3.69) post-bariatric surgery (n = 293) compared to general population controls (n = 656,353) [23].

LGA and macrosomia were investigated in 22 studies, and 21 were eligible for meta-analysis [22,26,27,29–31,33,34,36–39,41–46,51–53]. The ORs of an LGA baby post-bariatric surgery were more than halved (0.42, 95% CI 0.34–0.54, p < 0.001) (S5 Fig). There was substantial evidence of heterogeneity ( $I^2 = 69.5\%$ , 95% CI 52.4–80.5, p < 0.001). Subgroup analyses by type of surgery identified that the 'RYGB or BPD' group was associated with the biggest decrease in odds of LGA (OR 0.24, 95% CI 0.14–0.41, p < 0.001), in comparison with 'all bariatric surgery' (OR 0.51, 95% CI 0.39–0.67, p < 0.001), and 'LAGB or SG', which was not significant (OR 0.59, 95% CI 0.30–1.14, p = 0.116) (Fig 4B). Heterogeneity did not decrease in these subgroup analyses. Meta-regression revealed that sample size was significantly contributing to heterogeneity (residual  $I^2 = 61.21$ , coefficient = 0.249, p = 0.031) (S3B Table). The study excluded from the meta-analysis reported an adjusted OR of 0.03 (95% CI 0.01–0.21) for LGA post-bariatric surgery (n = 293) compared to general population controls (n = 656,353) [23].

Birth weight mean and standard deviation for babies born after maternal bariatric surgery and controls were reported in 17 studies [9,22–24,27,29,32,33,36,38,40,41,44,48,49,51,52]. WMD was significantly lower post-bariatric surgery (WMD –242.42 g, 95% CI –307.43 g to –-177.40 g, p < 0.001) (S6 Fig). Heterogeneity was substantial ( $I^2 = 75.7\%$ , 95% CI 61.1–84.8, p < 0.001) but reduced after subgroup analyses by surgery type. RYGB resulted in the largest reduction in birth weight (WMD –226.10 g, 95% CI –273.43 g to –178.78 g, p < 0.001), compared with 'all bariatric surgery' (WMD –223.71 g, 95% CI –273.68 g to –173.74 g, p < 0.001), and 'LAGB', for which the reduction was not significant (WMD –135.14 g, 95% CI –289.17 g to 18.90 g, p = 0.086). One study investigated only BPD, for which the mean difference was –500 g (95% CI –570.85 g to –429.15 g, p < 0.001).

#### NICU admission

NICU admission was reported in nine studies with babies born post-bariatric surgery being significantly more likely to be admitted to NICU (OR 1.41, 95% CI 1.25–1.59, p < 0.001) (Fig 5) [25,27,31,33,38,39,42,46,51]. There was no evidence of heterogeneity ( $I^2 = 0.0\%$ , 95% CI 0.0–64.8, p = 0.808).

#### Publication bias and sensitivity analyses

There was no evidence of small study effects for any outcome except LGA (p = 0.021), which may signal publication bias (S7 Fig, S4 Table). A subset of studies reported both crude and adjusted data for the adverse perinatal outcomes, but when compared, there was little difference in size or direction of associations (S8 Fig). Sensitivity analyses revealed that the results were robust, with only small changes in pooled effect sizes when meta-analysis were repeated with one study excluded (S5 Table).

#### Discussion

This systematic review and meta-analysis has demonstrated that perinatal mortality, congenital anomalies, preterm birth, SGA, and NICU admission are associated with increased odds in women who have had bariatric surgery prior to pregnancy compared to women without prior bariatric surgery. Postterm birth and LGA, however, are associated with decreased odds after bariatric surgery. Malabsorptive procedures were associated with a significant increase in SGA and decrease in LGA, whereas restrictive procedures were not. Subgrouping by type of surgery significantly reduced heterogeneity for the outcomes with a high *I*<sup>2</sup> value, whereas

NICU admission	Control group	Surgery (n/N)	Control (n/N)	OR (95% CI)
All bariatric surgery				C
Lesko and Peaceman, 2012	ppBMI Matched	8/70	8/140	2.13 (0.76, 5.94)
Chevrot et al. 2016	Obesity	12/139	6/139	• 2.09 (0.76, 5.75)
Parent et al. 2017	General Population	282/1859	956/8437	► 1.40 (1.21, 1.62)
RYGB				1 T
Wax et al. 2008	General Population	8/38	14/76	1.18 (0.45, 3.12)
Berlac et al. 2014	ppBMI Matched	83/415	137/827	1.26 (0.93, 1.70)
Feichtinger et al. 2016	ppBMI Matched	7/76	5/76	1.44 (0.44, 4.76)
Hammeken et al. 2017	ppBMI Matched	19/151	11/151	1.83 (0.84, 4.00)
LAGB or SG*				
Lapolla et al. 2010	Obesity	17/83	12/120	• 2.32 (1.04, 5.16)
Rottenstreich et al. 2018*	Obesity	4/119	5/119	0.79 (0.21, 3.03)
				1
Overall (I-squared = 0.0%, p = 0	.808)		<	1.41 (1.25, 1.59)
NOTE: Weights are from random	effects analysis			

**Fig 5. NICU admission after bariatric surgery meta-analysis.** Association between maternal bariatric surgery and NICU admission. Studies are presented as Author, year. The forest plot is stratified by type of surgery. n = cases of NICU admission. N = total group size. \*SG only. CI, confidence interval; LAGB, laparoscopic adjustable gastric banding; NICU, neonatal intensive care unit; OR, odds ratio; ppBMI, prepregnancy BMI matched; RYGB, Roux-en-Y gastric bypass; SG, sleeve gastrectomy.

https://doi.org/10.1371/journal.pmed.1002866.g005

subgrouping by control group did not. There was no evidence of publication bias for any outcome except LGA.

The increase in adverse perinatal outcomes could be related to malnutrition. Unlike restrictive procedures, which reduce stomach size and appetite, malabsorptive procedures bypass a portion of the small intestine where many vitamins and minerals are absorbed, making these patients particularly susceptible to nutrient deficiencies that may negatively affect a subsequent pregnancy [54]. The association between folic acid intake and neural tube defects is well established, and there are links between iron deficiency and preterm birth and between calcium and birth weight [55–57]. Impaired nutrient transport across the placenta is also associated with perinatal morbidity; however, there is limited evidence regarding placental function after bariatric surgery. The studies reporting data on congenital anomalies in pregnancy with and without prior bariatric surgery did not subgroup by type of anomaly—this would be valuable for future research to pinpoint the mechanism behind the anomalies. Another factor that may explain the increase in SGA infants is the increased glycaemic variability and postprandial hypoglycaemia observed after RYGB, as fetal growth has been found to be associated with maternal glucose nadir levels during oral glucose tolerance testing in pregnancy [58].

The strengths of this systematic review and meta-analysis include the thorough search strategy of multiple databases and supplementing this with hand searches of reference lists, citations, and relevant journals. All screening, data extraction, and quality assessment was carried out in duplicate to minimise human error. There are no randomised controlled trials, because of the nature of this research question, but all included studies were medium- to high-quality observational studies. This is the first meta-analysis, to our knowledge, to report significantly increased odds of perinatal mortality and congenital anomalies after bariatric surgery. This is also the first meta-analysis, to our knowledge, to investigate postterm birth after bariatric surgery, for which a significant decrease was found. The results for SGA, LGA, preterm birth, and NICU admission confirm the findings of past meta-analyses but with stronger associations than previously reported and the inclusion of 12 additional studies [9,22,24,27,28,32,33,40,42,46,49,50].

The results from our study are limited by the small sample sizes of some of the included studies. Multiple studies reported few, or even zero, cases of perinatal mortality or congenital anomalies and have therefore resulted in large CIs. Larger epidemiological studies or individual patient data (IPD) meta-analyses need to be carried out for this rare exposure and rare outcome combination. Additionally, there are no large studies exploring congenital anomalies and perinatal mortality specifically after restrictive surgery such as LAGB or SG, which may not have a detrimental effect. A number of studies have reported several adverse perinatal outcomes, many of which are linked, which may result in a loss of statistical and clinical independence. We were unable to include non–English language studies, and one non–English language study meeting our inclusion criteria was excluded. This study from France identified a significant decrease in macrosomia, as our meta-analysis did; however, it also found a decrease in SGA in contrast to the significant increase we found [59].

Women that become pregnant post-bariatric surgery tend to be older than the general population of pregnant women [7]. Many women also still have a BMI  $> 30 \text{ kg/m}^2$  despite the weight loss from surgery [45]. There is also evidence that alcohol use and smoking are increased after bariatric surgery [60]. The combination of increased maternal age, high BMI, and unhealthy behaviours in women after bariatric surgery plays a role in the development of adverse perinatal outcomes, in addition to the malnutrition. These are important confounders to consider when investigating perinatal outcomes in this group. When comparing ORs with adjustments made for these factors to unadjusted ORs, we did not see a change in the results. However, in a clinical setting, these factors and behaviours are important for the healthcare provider to take into account because of the evidence of the link with adverse perinatal outcomes. As with all meta-analyses of observational data, unmeasured confounding in the included studies may have implications on the results. Gestational weight gain (GWG) is another factor associated with perinatal outcomes such as birth weight; however, further research is required to determine how the relationship between GWG and pregnancy outcomes differs for women after bariatric surgery and whether current GWG guidelines can apply to this population.

The LAGB subgroup analyses tended to have larger CIs than any other subgroup. This may be due to smaller sample sizes or differences in LAGB band management. Some clinics actively manage gastric bands during pregnancy by deflating in cases of nausea or vomiting and inflating in cases of excess GWG [29]. Future studies should explore how band management could be used to achieve optimal pregnancy outcomes. The studies that combined all types of bariatric surgery drastically differed in surgery type composition, with studies reporting from 13.3% RYGB to 98% RYGB in their cohorts. It would be useful for future studies to separate outcomes by type of surgery or to conduct IPD meta-analyses on the existing data, which would enable standardisation of categories across studies.

Future studies should explore the effect of time to conception after different types of bariatric surgery, especially considering gestational weight loss and advanced maternal age. Many women that are previously considered to be infertile experience increased fertility after bariatric surgery, which may result in unexpected pregnancies immediately after surgery in the rapid weight loss phase [61]. Many clinics recommend waiting 12–18 months to conceive post-surgery, but the evidence base is limited for this.

Bariatric surgery prior to pregnancy is promising for reducing obesity-related comorbidities for the mother, and benefits include reduced risks of gestational diabetes and preeclampsia, which are both serious complications associated with adverse maternal and fetal outcomes. Our meta-analysis has shown that the risks of postterm birth and LGA babies are reduced after bariatric surgery; however, we have also identified adverse outcomes for the baby and efforts now need to be focused on how to reduce these. Internationally, guidelines exist for a variety of high-risk pregnancy groups such as those with diabetes, hypertension, and obesity. This study confirms that bariatric surgery patients that become pregnant are also a high-risk group, and guidelines for health professionals need to be developed as obesity and bariatric surgery increases. The current evidence base could be used to inform risk communication about potential future pregnancies with women of reproductive age prior to surgery. For women with a history of bariatric surgery, preconception nutritional support should be offered, and increased fetal, nutrition, and glucose monitoring is required throughout pregnancy. Further studies are required to determine whether restrictive surgery results in better perinatal outcomes than malabsorptive surgery without compromising maternal outcomes, and if so, these may be the preferred surgery for women of reproductive age.

#### Supporting information

**S1** Table. Search strategy for electronic databases and e-journals. (DOCX)

**S2** Table. Quality assessment scores for included studies. (DOCX)

**S3** Table. Meta-regression for outcomes with significant heterogeneity between studies. (DOCX)

**S4** Table. Eggers test of publication bias for perinatal outcomes after bariatric surgery. (DOCX)

**S5** Table. Sensitivity analyses for perinatal outcomes after bariatric surgery. (DOCX)

**S1 PRISMA** Checklist. PRISMA checklist for systematic reviews and meta-analyses. (DOC)

S1 Fig. Adapted Newcastle-Ottawa quality assessment scale for cohort studies. (DOCX)

**S2** Fig. Preterm birth after bariatric surgery meta-analysis. (DOCX)

S3 Fig. Gestational age (weeks) after bariatric surgery meta-analysis with subtotals by type of surgery.

(DOCX)

**S4** Fig. Small for gestational age after bariatric surgery meta-analysis. (DOCX)

**S5 Fig. Large for gestational age after bariatric surgery meta-analysis.** (DOCX)

S6 Fig. Birth weight (grams) after bariatric surgery meta-analysis with subtotals by type of surgery.

(DOCX)

**S7** Fig. Funnel plots of publication bias for perinatal outcomes after bariatric surgery. (DOCX)

**S8** Fig. Crude versus adjusted data for studies reporting adjusted odds ratios. (DOCX)

#### Acknowledgments

We would like to thank the authors that responded to email requests, Dr H. Salihu and Dr O. Stephansson.

#### **Author Contributions**

Conceptualization: Zainab Akhter, Judith Rankin, Roland Devlieger, Nicola Heslehurst.

Data curation: Zainab Akhter, Judith Rankin, Dries Ceulemans, Lem Ngongalah, Roger Ackroyd, Nicola Heslehurst.

Formal analysis: Zainab Akhter.

Funding acquisition: Judith Rankin, Nicola Heslehurst.

Investigation: Zainab Akhter.

Methodology: Zainab Akhter, Judith Rankin, Dries Ceulemans, Lem Ngongalah, Rute Vieira, Nicola Heslehurst.

Project administration: Zainab Akhter.

Supervision: Judith Rankin, Roger Ackroyd, Roland Devlieger, Nicola Heslehurst.

- Validation: Judith Rankin, Dries Ceulemans, Lem Ngongalah, Roger Ackroyd, Roland Devlieger, Rute Vieira, Nicola Heslehurst.
- Writing original draft: Zainab Akhter.
- Writing review & editing: Zainab Akhter, Judith Rankin, Dries Ceulemans, Lem Ngongalah, Roger Ackroyd, Roland Devlieger, Rute Vieira, Nicola Heslehurst.

#### References

- 1. World Health Organization. Obesity and Overweight: Fact sheet [Internet] [cited 2019 Jan 9]. <u>https://www.who.int/en/news-room/fact-sheets/detail/obesity-and-overweight</u>
- Catalano PM, Shankar K. Obesity and pregnancy: mechanisms of short term and long term adverse consequences for mother and child. BMJ. 2017; 356: j1. <u>https://doi.org/10.1136/bmj.j1</u> PMID: <u>28179267</u>
- Marchi J, Berg M, Dencker A, Olander EK, Begley C. Risks associated with obesity in pregnancy, for the mother and baby: a systematic review of reviews. Obes Rev. 2015; 16: 621–638. <u>https://doi.org/10. 1111/obr.12288</u> PMID: <u>26016557</u>
- Heslehurst N, Vieira R, Hayes L, Crowe L, Jones D, Robalino S, et al. Maternal body mass index and post-term birth: a systematic review and meta-analysis. Obes Rev. 2017; 18: 293–308. <u>https://doi.org/ 10.1111/obr.12489</u> PMID: <u>28085991</u>
- Godfrey KM, Reynolds RM, Prescott SL, Nyirenda M, Jaddoe VW, Eriksson JG, et al. Influence of maternal obesity on the long-term health of offspring. Lancet Diabetes Endocrinol. 2017; 5: 53–64. <u>https://doi.org/10.1016/S2213-8587(16)30107-3</u> PMID: <u>27743978</u>

- Ovrebo B, Strommen M, Kulseng B, Martins C. Bariatric surgery versus lifestyle interventions for severe obesity: 5-year changes in body weight, risk factors and comorbidities. Clin Obes. 2017; 7: 183–190. <u>https://doi.org/10.1111/cob.12190</u> PMID: <u>28320081</u>
- Edison E, Whyte M, van Vlymen J, Jones S, Gatenby P, de Lusignan S, et al. Bariatric Surgery in Obese Women of Reproductive Age Improves Conditions That Underlie Fertility and Pregnancy Outcomes: Retrospective Cohort Study of UK National Bariatric Surgery Registry (NBSR). Obes Surg. 2016; 26: 2837–2842. https://doi.org/10.1007/s11695-016-2202-4 PMID: 27317009
- Yi XY, Li QF, Zhang J, Wang ZH. A meta-analysis of maternal and fetal outcomes of pregnancy after bariatric surgery. Int J Gynaecol Obstet. 2015; 130: 3–9. <u>https://doi.org/10.1016/j.ijgo.2015.01.011</u> PMID: <u>25863541</u>
- Gascoin G, Gerard M, Sallé A, Becouarn G, Rouleau S, Sentilhes L, et al. Risk of low birth weight and micronutrient deficiencies in neonates from mothers after gastric bypass: a case control study. Surg Obes Relat Dis. 2017; 13: 1384–1391. https://doi.org/10.1016/j.soard.2017.03.017 PMID: 28526433
- Stephenson J, Heslehurst N, Hall J, Schoenaker DAJM, Hutchinson J, Cade JE, et al. Before the beginning: nutrition and lifestyle in the preconception period and its importance for future health. Lancet. 2018; 391: 1830–1841. https://doi.org/10.1016/S0140-6736(18)30311-8 PMID: 29673873
- O'Kane M, Pinkney J, Aasheim E, Barth J, Batterham R, Welbourn R. BOMSS GP Guidance: Management of nutrition following bariatric surgery [Internet]. August 2014 [cited 2018 Dec 6]. <u>http://www.bomss.org.uk/wp-content/uploads/2014/09/GP\_Guidance-Final-version-1Oct141.pdf</u>.
- Jans G, Matthys C, Bogaerts A, Lannoo M, Verhaeghe J, Van der Schueren B, et al. Maternal micronutrient deficiencies and related adverse neonatal outcomes after bariatric surgery: a systematic review. Adv Nutr. 2015; 6: 420–429. <u>https://doi.org/10.3945/an.114.008086</u> PMID: <u>26178026</u>
- Kwong W, Tomlinson G, Feig DS. Maternal and neonatal outcomes after bariatric surgery; a systematic review and meta-analysis: do the benefits outweigh the risks? Am J Obstet Gynecol. 2018; 218: 573– 580. <u>https://doi.org/10.1016/j.ajog.2018.02.003</u> PMID: 29454871
- Galazis N, Docheva N, Simillis C, Nicolaides KH. Maternal and neonatal outcomes in women undergoing bariatric surgery: a systematic review and meta-analysis. Eur J Obstet Gynecol Reprod Biol. 2014; 181: 45–53. <u>https://doi.org/10.1016/j.ejogrb.2014.07.015</u> PMID: <u>25126981</u>
- Moher D, Liberati A, Tetzlaff J, Altman DG, The PRISMA GROUP. Preferred Reporting Items for Systematic Reviews and Meta-Analyses: The PRISMA Statement. PLoS Med. 2009; 6(7): e1000097. <u>https://doi.org/10.1371/journal.pmed.1000097</u> PMID: <u>19621072</u>
- Stroup DF, Berlin JA, Morton SC, Olkin I, Williamson GD, Rennie D, et al. Meta-analysis of observational studies in epidemiology: a proposal for reporting. Meta-analysis Of Observational Studies in Epidemiology (MOOSE) group. JAMA. 2000; 283: 2008–2012. <u>https://doi.org/10.1001/jama.283.15.2008</u> PMID: <u>10789670</u>
- 17. McHugh ML. Interrater reliability: the kappa statistic. Biochem Med 2012; 22: 276–282.
- Smith J, Cianflone K, Biron S, Hould FS, Lebel S, Marceau S, et al. Effects of maternal surgical weight loss in mothers on intergenerational transmission of obesity. J Clin Endocrinol Metab. 2009; 94: 4275– 4283. <u>https://doi.org/10.1210/jc.2009-0709</u> PMID: <u>19820018</u>
- Dell'Agnolo CM, Cyr C, de Montigny F, de Barros Carvalho MD, Pelloso SM. Pregnancy after Bariatric Surgery: Obstetric and Perinatal Outcomes and the Growth and Development of Children. Obes Surg. 2015; 25: 2030–2039. <u>https://doi.org/10.1007/s11695-015-1668-9</u> PMID: <u>25893647</u>
- Sheiner E, Levy A, Silverberg D, Menes TS, Levy I, Katz M, et al. Pregnancy after bariatric surgery is not associated with adverse perinatal outcome. Am J Obstet Gynecol. 2004; 190: 1335–1340. <u>https:// doi.org/10.1016/j.ajog.2003.11.004</u> PMID: <u>15167839</u>
- Santulli P, Mandelbrot L, Facchiano E, Dussaux C, Ceccaldi PF, Ledoux S, et al. Obstetrical and neonatal outcomes of pregnancies following gastric bypass surgery: a retrospective cohort study in a French referral centre. Obes Surg.2010; 20: 1501–1508. <u>https://doi.org/10.1007/s11695-010-0260-6</u> PMID: 20803358
- Adams TD, Hammoud AO, Davidson LE, Laferrère B, Fraser A, Stanford JB, et al. Maternal and neonatal outcomes for pregnancies before and after gastric bypass surgery. Int J Obes (Lond). 2015; 39: 686–694.
- 23. Belogolovkin V, Salihu HM, Weldeselasse H, Biroscak BJ, August EM, Mbah AK, et al. Impact of prior bariatric surgery on maternal and fetal outcomes among obese and non-obese mothers. Arch Gynecol Obstet. 2012; 285: 1211–1218. <u>https://doi.org/10.1007/s00404-011-2134-0</u> PMID: <u>22057892</u>
- Berglind D, Willmer M, Naslund E, Tynelius P, Sorensen TI, Rasmussen F. Differences in gestational weight gain between pregnancies before and after maternal bariatric surgery correlate with differences in birth weight but not with scores on the body mass index in early childhood. Pediatr Obes. 2014; 9: 427–434. https://doi.org/10.1111/j.2047-6310.2013.00205.x PMID: 24339139

- Berlac JF, Skovlund CW, Lidegaard O. Obstetrical and neonatal outcomes in women following gastric bypass: a Danish national cohort study. Acta Obstet Gynecol Scand. 2014; 93: 447–453. <u>https://doi. org/10.1111/aogs.12368</u> PMID: <u>24592873</u>
- Burke AE, Bennett WL, Jamshidi RM, Gilson MM, Clark JM, Segal JB, et al. Reduced incidence of gestational diabetes with bariatric surgery. J Am Coll Surg. 2010; 211: 169–175. <u>https://doi.org/10.1016/j.</u> jamcollsurg.2010.03.029 PMID: 20670854
- Chevrot A, Kayem G, Coupaye M, Lesage N, Msika S, Mandelbrot L. Impact of bariatric surgery on fetal growth restriction: Experience of a perinatal and bariatric surgery center. Am J Obstet Gynecol. 2016; 214: 655.e1–7
- Dell'Agnolo CM, Carvalho MD, Pelloso SM. Pregnancy after bariatric surgery: implications for mother and newborn. Obes Surg. 2011; 21: 699–706. <u>https://doi.org/10.1007/s11695-011-0363-8</u> PMID: 21331506
- Dixon JB, Dixon ME, O'Brien PE. Birth outcomes in obese women after laparoscopic adjustable gastric banding. Obstet Gynecol. 2005; 106: 965–972. <u>https://doi.org/10.1097/01.AOG.0000181821.82022.82</u> PMID: <u>16260513</u>
- Ducarme G, Revaux A, Rodrigues A, Aissaoui F, Pharisien I, Uzan M. Obstetric outcome following laparoscopic adjustable gastric banding. Int J Gynaecol Obstet. 2007; 98: 244–247. <u>https://doi.org/10. 1016/j.ijgo.2007.02.020</u> PMID: <u>17433814</u>
- Feichtinger M, Stopp T, Hofmann S, Springer S, Pils S, Kautzy-WIller A, et al. Altered glucose profiles and risk for hypoglycaemia during oral glucose tolerance testing in pregnancies after gastric bypass surgery. Diabetologia. 2017; 60: 153–157. <u>https://doi.org/10.1007/s00125-016-4128-8</u> PMID: <u>27757488</u>
- Goldman RH, Missmer SA, Robinson MK, Farland LV, Ginsburg ES. Reproductive Outcomes Differ Following Roux-en-Y Gastric Bypass and Adjustable Gastric Band Compared with Those of an Obese Non-Surgical Group. Obes Surg. 2016; 26: 2581–2589. <u>https://doi.org/10.1007/s11695-016-2158-4</u> PMID: 27052316
- 33. Hammeken LH, Betsagoo R, Jensen AN, Sørensen AN, Overgaard C. Nutrient deficiency and obstetrical outcomes in pregnant women following Roux-en-Y gastric bypass: A retrospective Danish cohort study with a matched comparison group. Eur J Obstet Gynecol Reprod Biol. 2017; 216: 56–60. <u>https://doi.org/10.1016/j.ejogrb.2017.07.016</u> PMID: 28732251
- Johansson K, Cnattingius S, Näslund I, Roos N, Trolle Lagerros Y, Granath F, et al. Outcomes of pregnancy after bariatric surgery. N Engl J Med. 2015; 372: 814–824. <u>https://doi.org/10.1056/ NEJMoa1405789</u> PMID: 25714159
- **35.** Josefsson A, Bladh M, Wirehn AB, Sydsjo G. Risk for congenital malformations in offspring of women who have undergone bariatric surgery. A national cohort. BJOG. 2013; 120: 1477–1482. <u>https://doi.org/10.1111/1471-0528.12365</u> PMID: 23927006
- Josefsson A, Blomberg M, Bladh M, Frederiksen SG, Sydsjo G. Bariatric surgery in a national cohort of women: sociodemographics and obstetric outcomes. Am J Obstet Gynecol 2011; 205: 206.e201–208.
- Kjaer MM, Lauenborg J, Breum BM, Nilas L. The risk of adverse pregnancy outcome after bariatric surgery: a nationwide register-based matched cohort study. Am J Obstet Gynecol. 2013; 208: 464.e461– 465.
- Lapolla A, Marangon M, Dalfrà MG, Segato G, De Luca M, Fedele D, et al. Pregnancy outcome in morbidly obese women before and after laparoscopic gastric banding. Obes Surg. 2010; 20: 1251–1257. https://doi.org/10.1007/s11695-010-0199-7 PMID: 20524157
- Lesko J, Peaceman A. Pregnancy outcomes in women after bariatric surgery compared with obese and morbidly obese controls. Obstet Gynecol. 2012; 119: 547–554. <u>https://doi.org/10.1097/AOG.</u> <u>0b013e318239060e</u> PMID: <u>22353952</u>
- Machado S, Pereira S, Saboya C, Saunders C, Ramalho A. Influence Of Roux-En-Y Gastric Bypass On Obstetric And Perinatal Outcomes: A Comparative Study. Obes Open Access. 2017; 3.
- Marceau P, Kaufman D, Biron S, Hould FS, Lebel S, Marceau S, et al. Outcome of pregnancies after biliopancreatic diversion. Obes Surg. 2004; 14: 318–324. <u>https://doi.org/10.1381/ 096089204322917819</u> PMID: <u>15072650</u>
- Parent B, Martopullo I, Weiss NS, Khandelwal S, Fay EE, Rowhani-Rahbar A. Bariatric surgery in women of childbearing age, timing between an operation and birth, and associated perinatal complications. JAMA Surg. 2017; 152: 128–135. https://doi.org/10.1001/jamasurg.2016.3621 PMID: 27760265
- 43. Parker MH, Berghella V, Nijjar JB. Bariatric surgery and associated adverse pregnancy outcomes among obese women. J Matern Fetal Neonatal Med. 2016; 29: 1747–1750. <u>https://doi.org/10.3109/ 14767058.2015.1060214</u> PMID: 26217944

- Patel JA, Patel NA, Thomas RL, Nelms JK, Colella JJ. Pregnancy outcomes after laparoscopic Rouxen-Y gastric bypass. Surg Obes Relat Dis. 2008; 4: 39–45. <u>https://doi.org/10.1016/j.soard.2007.10.008</u> PMID: <u>18201669</u>
- 45. Roos N, Neovius M, Cnattingius S, Trolle Lagerros Y, Sääf M, Granath F, et al. Perinatal outcomes after bariatric surgery: nationwide population based matched cohort study. BMJ. 2013; 347: f6460. <u>https://</u> doi.org/10.1136/bmj.f6460 PMID: 24222480
- 46. Rottenstreich A, Elchalal U, Kleinstern G, Beglaibter N, Khalaileh A, Elazary R. Maternal and Perinatal Outcomes After Laparoscopic Sleeve Gastrectomy. Obstet Gynecol. 2018; 131: 451–456. <u>https://doi.org/10.1097/AOG.0000000002481</u> PMID: 29420411
- 47. Shai D, Shoham-Vardi I, Amsalem D, Silverberg D, Levi I, Sheiner E. Pregnancy outcome of patients following bariatric surgery as compared with obese women: a population-based study. J Matern Fetal Neonatal Med. 2014; 27: 275–278. <u>https://doi.org/10.3109/14767058.2013.814632</u> PMID: <u>23773032</u>
- 48. Skull AJ, Slater GH, Duncombe JE, Fielding GA. Laparoscopic adjustable banding in pregnancy: safety, patient tolerance and effect on obesity-related pregnancy outcomes. Obes Surg. 2004; 14: 230–235. https://doi.org/10.1381/096089204322857618 PMID: 15018752
- 49. Stentebjerg LL, Andersen LLT, Renault K, Stoving RK, Jensen DM. Pregnancy and perinatal outcomes according to surgery to conception interval and gestational weight gain in women with previous gastric bypass. J Matern Fetal Neonatal Med. 2017; 30: 1182–1188. <u>https://doi.org/10.1080/14767058.2016.1208746</u> PMID: 27426696
- Stephansson O, Johansson K, Söderling J, Näslund I, Neovius M. Delivery outcomes in term births after bariatric surgery: Population-based matched cohort study. PLoS Med. 2018; 15(9): e1002656 https://doi.org/10.1371/journal.pmed.1002656 PMID: 30256796
- Wax JR, Cartin A, Wolff R, Lepich S, Pinette MG, Blackstone J. Pregnancy following gastric bypass surgery for morbid obesity: maternal and neonatal outcomes. Obes Surg. 2008; 18: 540–544. <u>https://doi.org/10.1007/s11695-008-9459-1</u> PMID: 18317852
- Weintraub AY, Levy A, Levi I, Mazor M, Wiznitzer A, Sheiner E. Effect of bariatric surgery on pregnancy outcome. Int J Gynaecol Obstet. 2008; 103: 246–251. <u>https://doi.org/10.1016/j.ijgo.2008.07.008</u> PMID: 18768177
- Wittgrove AC, Jester L, Wittgrove P, Clark GW. Pregnancy following gastric bypass for morbid obesity. Obes Surg. 1998; 8: 461–466. https://doi.org/10.1381/096089298765554368 PMID: 9731683
- Shankar P, Boylan M, Sriram K. Micronutrient deficiencies after bariatric surgery. Nutrition. 2010; 26: 1031–1037. <u>https://doi.org/10.1016/j.nut.2009.12.003</u> PMID: <u>20363593</u>
- Czeizel AE, Dudas I. Prevention of the first occurrence of neural-tube defects by periconceptional vitamin supplementation. N Engl J Med. 1992; 327: 1832–1835. <u>https://doi.org/10.1056/</u> NEJM199212243272602 PMID: 1307234
- Little MP, Brocard P, Elliott P, Steer PJ. Hemoglobin concentration in pregnancy and perinatal mortality: A London-based cohort study. Am J Obstet Gynecol. 2005; 193: 220–226. <u>https://doi.org/10.1016/j.ajog.2004.11.053</u> PMID: 16021083
- Mannion CA, Gray-Donald K, Koski KG. Association of low intake of milk and vitamin D during pregnancy with decreased birth weight. CMAJ. 2006; 174: 1273–1277. <u>https://doi.org/10.1503/cmaj.</u> 1041388 PMID: 16636326
- Göbl CS, Bozkurt L, Tura A, Leutner M, Andrei L, Fahr L, et al. Assessment of glucose regulation in pregnancy after gastric bypass surgery. Diabetologia. 2017; 60: 2504–2513. <u>https://doi.org/10.1007/</u> s00125-017-4437-6 PMID: 28918470
- Fumery L, Pigeyre M, Fournier C, Arnalsteen L, Rivaux G, Subtil D, et al. Impact de la chirurgie bariatrique sur le pronostic obstétrical. Gynécologie Obstétrique & Fertilité. 2013; 41: 156–163.
- Conason A, Teixeira J, Hsu C, Puma L, Knafo D, Geliebter A. Substance use following bariatric weight loss surgery. JAMA Surg. 2013; 148:145–150. <u>https://doi.org/10.1001/2013.jamasurg.265</u> PMID: 23560285
- Milone M, De Placido G, Musella M, Sosa Fernandez LM, Sosa Fernandez LV, Campana G, et al. Incidence of Successful Pregnancy After Weight Loss Interventions in Infertile Women: a Systematic Review and Meta-Analysis of the Literature. Obes Surg. 2016; 26: 443–451. <u>https://doi.org/10.1007/s11695-015-1998-7</u> PMID: <u>26661108</u>